Author's response to reviews

Title: A Clinical Pathway for Community-Acquired Pneumonia: An Observational Cohort Study

Authors:

Christopher R Frei (freic@uthscsa.edu)
Allison M Bell (bella3@uthscsa.edu)
Kristi A Traugott (ktraugott@hotmail.com)
Terry C Jaso (tjaso@seton.org)
Kelly R Daniels (kdaniels@mail.utexas.edu)
Eric M Mortensen (mortensene@uthscsa.edu)
Marcos I Restrepo (restrepom@uthscsa.edu)
Christine U Oramasionwu (chrorams@mail.utexas.edu)
Andres D Ruiz (oakdellwellness@gmail.com)
William R Mylchreest (wmylchreest@seton.org)
Vanja Sikirica (vanja_sikirica@yahoo.com)
Monika R Raut (mraut1@its.jnj.com)
Alan Fisher (acf900@gmail.com)
Jeff R Schein (jschein@its.jnj.com)

Version: 7 Date: 25 June 2011

Author's response to reviews:

Minor edits were made to these page proofs. These include changes to author affiliations, spelling, and grammar.